Author's response to reviews

Title: Drug related problems in type 2 diabetes patients with hypertension

Authors:

Hasniza Zaman Huri (hasnizazh@um.edu.my)
Hoo Fun Wee (hoofunwee@hotmail.com)

Version: 2 Date: 22 August 2012

Author's response to reviews:

The statement of ethical committee approval and Declaration of Helsinki have been included under subtitle of 'study population'

The section of conflict of interest has been included after the conclusion part